Author's response to reviews

Title: Prehospital triage accuracy in a Criteria Based Dispatch centre

Authors:

fabrice dami (fabrice.dami@chuv.ch)
christel golay (christel.golay@unil.ch)
mathieu pasquier (mathieu.pasquier@chuv.ch)
pierre-nicolas carron (pierre-nicolas.carron@chuv.ch)
vincent fuchs (vincent.fuchs@urgences-sante.ch)
olivier hugli (olivier.hugli@chuv.ch)

Version: 5
Date: 6 October 2015

Author's response to reviews:

Editor's Comments:

1. Include all author e-mails on the title page
   INCLUDED
2. Include Abbrev section
   INCLUDED
3. List figure legends after the References
   INCLUDED